Author’s response to reviews

Title: Validation of the 8th Edition UICC/AJCC TNM staging system for HPV associated oropharyngeal cancer patients managed with Contemporary Chemo-Radiotherapy

Authors:

Kirsten van Gysen (kirstyvangysen@gmail.com)
Mark Stevens (mjstevens@ausdoctors.net)
Linxin Guo (Lesley.guo@health.nsw.gov.au)
Dasantha Jayamanne (Dasantha.jayamanne@health.nsw.gov.au)
David Veivers (Drveivers@nshank.com.au)
Andrew Wignall (drwignall@wignallent.com.au)
Leo Pang (drleopang@gmail.com)
Alexander Guminski (alexander.guminski@health.nsw.gov.au)
Adrian Lee (Adrian.lee@sydney.edu.au)
George Hruby (George.hruby@health.nsw.gov.au)
Paula Macleod (Paula.macleod@health.nsw.gov.au)
Alon Taylor (Atay5510@uni.sydney.edu.au)
Thomas Eade (Thomas.Eade@health.nsw.gov.au)

Version: 1 Date: 26 Jun 2019

Author’s response to reviews:

1. Written consent was obtained and this had been included in the section 'Ethics Approval and Consent to Participate'

2. Andrew Wignall has been included in the authors' contributions section stating his individual contributions.

3. A separate heading of 'Figure Legends' has been included after the References.
4. The Response to reviewers and the cover letter have been removed from the submissions supplementary files.

5. The final clean manuscript has been uploaded. Figures have been uploaded as separate files.

Thank you